Author’s response to reviews

Title: Influence of oral premedication and prewarming on core temperature of cardiac surgical patients: A prospective, randomized, controlled trial

Authors:
Anselm Bräuer (abraeue@gwdg.de)
Michaela Müller (michaela.mueller@med.uni-goettingen.de)
Anna Wetz (anna.wetz@med.uni-goettingen.de)
Michael Quintel (mquintel@med.uni-goettingen.de)
Ivo Brandes (ibrandes@med.uni-goettingen.de;ibrande@gwdg.de)

Version: 3 Date: 06 Mar 2019

Author’s response to reviews:

Dear Amy Branch-Hollis,

thank you very much for your email regarding the manuscript BANE-D-18-00408R1 "Influence of oral premedication and prewarming on core temperature of cardiac surgical patients: A prospective, randomized, controlled trial". Enclosed in this current re-submission please find the manuscript with the requested changes. Please note that Fig.3 now contains the correct value for \( r^2 \) (formerly \( r^2 = 0.015 \), now corrected to \( r^2 = 0.15 \)).

The order of the authors has been corrected, there has been a mistake during the first re-submission process.

If you have any further questions please do not hesitate to contact me. I am happy to be of any assistance.

Yours sincerely,

Ivo F. Brandes